Author’s response to reviews

Title: Relationship between ALDH2 Genotype and In-stent Restenosis in Chinese Han Patients after Percutaneous Coronary Intervention

Authors:

Lizhi Lv (provlvizhi@126.com)
Weijie Ye (463111153@qq.com)
Peiyuan Song (song_ng_peiyuan@126.com)
Yubin Chen (470643668@qq.com)
Jing Yang (fccyangj3@zzu.edu.cn)
Congmin Zhang (fcczhangcm@zzu.edu.cn)
Xiaoping Chen (chenxp74@hotmail.com)
Fanyan Luo (drlfy@163.com)

Version: 7 Date: 01 Jun 2019

Author’s response to reviews:

Dear Editor:

We would like to thank the editor for giving us a chance to resubmit our paper entitled “Relationship between Aldehyde ALDH2 Genotype and In-Stent Restenosis in Chinese Han Patients after Percutaneous Coronary Intervention” (BCAR-D-18-00746). I hope this revision can meet your requirements and be accepted as soon as possible. I need this article to complete my graduation.
Question 1. Please add the limitation into the limitation section, point 1 and consider to revise the statement "…First, the sample size was small. Moreover, for analyzing the changes in percentage diameter stenosis as a continuous variable, we had to exclude those patients with a reduction of less than 50% in in-stent luminal diameter, because the angiographic images were not re-accessible for a technical reason. A prospective study with a larger sample size is required to validate our findings…." 

Answer: Thank you for your advice. We added the limitation into the limitation section (lines 335-340).

Question 2. For a reference citation in text with an author name, please only list the author's family name. On page 20, lines 49-50, "Jiali Wang et al..." should be revised to "Wang et al...". Please also check this through the manuscript.

Answer: Thank you for your correction. We revised the manuscript and changed the author name (line 320).

Sincerely,

Lizhi Lv, M.D